Author’s response to reviews

Title: Knowledge, uptake of preconception care and associated factors among reproductive age group women in West Shewa zone, Ethiopia, 2018.

Authors:

Daniel Belema Fekene (danibelema@gmail.com)

BENYAM SEIFU WOLDEYES (benyamseifu77@gmail.com)

MARU MOSSISA ERENA (mossisamaru@gmail.com)

GETU ALEMU DEMISSE (getualemu103@gmail.com)

Version: 2 Date: 08 Aug 2019

Author’s response to reviews:

Dear editor, thank you for your comments and suggestion, we try to minimize textual overlap by rephrase. Concerning age of participant, there is no age category less than 18 in the study participant and it was corrected again in the ethical clearance.